\begin{thebibliography}{10}

\bibitem{DBLP:conf/pldi/Adl-TabatabaiLMMSS06}
A.-R. Adl-Tabatabai, B.~T. Lewis, V.~Menon, B.~R. Murphy, B.~Saha, and
  T.~Shpeisman.
\newblock Compiler and runtime support for efficient software transactional
  memory.
\newblock In {\em PLDI}, 2006.

\bibitem{ananian:utm:hpca:2005}
C.~S. Ananian, K.~Asanovic, B.~C. Kuszmaul, C.~E. Leiserson, and S.~Lie.
\newblock Unbounded transactional memory.
\newblock In {\em HPCA}, 2005.

\bibitem{blundell:deconstructing-transactions:wddd:2005}
C.~Blundell, E.~C. Lewis, and M.~M. Martin.
\newblock Deconstructing transactional semantics: The subtleties of atomicity.
\newblock In {\em WDDD}, 2005.

\bibitem{DBLP:conf/ccs/ChenW02}
H.~Chen and D.~Wagner.
\newblock Mops: an infrastructure for examining security properties of
  software.
\newblock In {\em CCS}, 2002.

\bibitem{CDL99}
T.~M. Chilimbi, B.~Davidson, and J.~R. Larus.
\newblock Cache-conscious structure definition.
\newblock In {\em PLDI}, 1999.

\bibitem{chung:common-case:hpca:2006}
J.~Chung, H.~Chafi, C.~Cao~Minh, A.~McDonald, B.~D. Carlstrom, C.~Kozyrakis, ,
  and K.~Olukotun.
\newblock The common case transactional behavior of multithreaded programs.
\newblock In {\em HPCA}. 2006.

\bibitem{ConchonFilliatreSignoles05}
S.~Conchon, J.-C. Filli\^atre, and J.~Signoles.
\newblock Le foncteur sonne toujours deux fois.
\newblock In {\em Seizi\`emes Journ\'ees Francophones des Langages
  Applicatifs}. INRIA, 2005.

\bibitem{DBLP:conf/pldi/CooperK88}
K.~D. Cooper and K.~Kennedy.
\newblock Interprocedural side-effect analysis in linear time.
\newblock In {\em PLDI}, 1988.

\bibitem{CLRS}
T.~H. Cormen, C.~E. Leiserson, R.~L. Rivest, and C.~Stein.
\newblock {\em Introduction to Algorithms}.
\newblock MIT Press, second edition, 2001.

\bibitem{POPL:CC77}
P.~Cousot and R.~Cousot.
\newblock Abstract interpretation: {A} unified lattice model for static
  analysis of programs by construction of approximation of fixed points.
\newblock In {\em POPL}, 1977.

\bibitem{IFIP:CC78}
P.~Cousot and R.~Cousot.
\newblock Static determination of dynamic properties of recursive procedures.
\newblock In {\em IFIP}, 1978.

\bibitem{damron:hytm:asplos:2006}
P.~Damron, A.~Fedorova, Y.~Lev, V.~Luchango, M.~Moir, and D.~Nussbaum.
\newblock Hybrid transactional memory.
\newblock In {\em ASPLOS}, 2006.

\bibitem{DBLP:conf/sosp/EnglerA03}
D.~R. Engler and K.~Ashcraft.
\newblock Racerx: effective, static detection of race conditions and deadlocks.
\newblock In {\em SOSP}, 2003.

\bibitem{DBLP:conf/osdi/EnglerCCH00}
D.~R. Engler, B.~Chelf, A.~Chou, and S.~Hallem.
\newblock Checking system rules using system-specific, programmer-written
  compiler extensions.
\newblock In {\em OSDI}, 2000.

\bibitem{hammond:tcc:isca:2004}
L.~Hammond, V.~Wong, M.~Chen, B.~D. Carlstrom, and J.~D. Davis.
\newblock Transactional memory coherence and consistency.
\newblock In {\em ISCA}, 2004.

\bibitem{DBLP:conf/oopsla/HarrisF03}
T.~Harris and K.~Fraser.
\newblock Language support for lightweight transactions.
\newblock In {\em OOPSLA}, 2003.

\bibitem{DBLP:conf/ppopp/HarrisMPH05}
T.~Harris, S.~Marlow, S.~L.~P. Jones, and M.~Herlihy.
\newblock Composable memory transactions.
\newblock In {\em PPoPP}, 2005.

\bibitem{DBLP:conf/pldi/HarrisPST06}
T.~Harris, M.~Plesko, A.~Shinnar, and D.~Tarditi.
\newblock Optimizing memory transactions.
\newblock In {\em PLDI}, 2006.

\bibitem{herlihy:sw-trans-mem:podc:2003}
M.~Herlihy, V.~Luchangco, M.~Moir, and W.~Scherer~III.
\newblock Software transactional memory for dynamic-sized data structures.
\newblock In {\em PODC}, 2003.

\bibitem{herlihy:transactional:isca:1993}
M.~Herlihy and J.~E.~B. Moss.
\newblock Transactional memory: Architectural support for lock-free data
  structures.
\newblock In {\em ISCA}, pages 289--300, 1993.

\bibitem{kumar:hybrid-tm:ppopp:2006}
S.~Kumar, M.~Chu, C.~J. Hughes, P.~Kundu, and A.~Nguyen.
\newblock Hybrid transactional memory.
\newblock In {\em PPoPP}, 2006.

\bibitem{buddy}
J.~Lind-Nielsen.
\newblock {BuDDy} - a binary decision diagram package, 1997.
\newblock http://sourceforge.net/projects/buddy.

\bibitem{martin:gems:can:2005}
M.~M. Martin, D.~J. Sorin, B.~M. Beckmann, M.~R. Marty, M.~Xu, A.~R.
  Alameldeen, K.~E. Moore, M.~D. Hill, and D.~A. Wood.
\newblock Multifacet's {General} {Execution-driven} {Multiprocessor}
  {Simulator} ({GEMS}) {Toolset}.
\newblock {\em Computer Architecture News}, 2005.

\bibitem{moore:logtm:hpca:2006}
K.~E. Moore, J.~Bobba, M.~J. Moravan, M.~D. Hill, and D.~A. Wood.
\newblock {LogTM}: Log-based transactional memory.
\newblock In {\em HPCA}, 2006.

\bibitem{DBLP:conf/cc/NeculaMRW02}
G.~C. Necula, S.~McPeak, S.~P. Rahul, and W.~Weimer.
\newblock {CIL}: Intermediate language and tools for analysis and
  transformation of c programs.
\newblock In {\em CC}, 2002.

\bibitem{pratikakis:locksmith:pldi:2006}
P.~Pratikakis, J.~S. Foster, and M.~Hicks.
\newblock {LOCKSMITH}: Context-sensitive correlation analysis for race
  detection.
\newblock In {\em PLDI}, 2006.

\bibitem{rajwar:tlr:asplos:2002}
R.~Rajwar and J.~R. Goodman.
\newblock Transactional lock-free execution of lock-based programs.
\newblock In {\em ASPLOS}, 2002.

\bibitem{rajwar:vtm:isca:2005}
R.~Rajwar, M.~Herlihy, and K.~Lai.
\newblock Virtualizing transactional memory.
\newblock In {\em ISCA}, 2005.

\bibitem{RSJM}
T.~Reps, S.~Schwoon, S.~Jha, and D.~Melski.
\newblock Weighted pushdown systems and their application to interprocedural
  dataflow analysis.
\newblock {\em SCP}, 58, 2005.

\bibitem{DBLP:journals/toplas/Reps00}
T.~W. Reps.
\newblock Undecidability of context-sensitive data-independence analysis.
\newblock {\em ACM Trans. Program. Lang. Syst.}, 22(1), 2000.

\bibitem{DBLP:conf/icfp/RingenburgG05}
M.~F. Ringenburg and D.~Grossman.
\newblock Atomcaml: first-class atomicity via rollback.
\newblock In {\em ICFP}, 2005.

\bibitem{kn:SRH96}
M.~Sagiv, T.~Reps, and S.~Horwitz.
\newblock Precise interprocedural dataflow analysis with applications to
  constant propagation.
\newblock {\em Theor.\ Comp.\ Sci.}, 167, 1996.

\bibitem{saha:mcrt:ppopp:2006}
B.~Saha, A.-R. Adl-Tabatabai, R.~L. Hudson, C.~C. Minh, and B.~Hertzberg.
\newblock Mcrt-stm: a high performance software transactional memory system for
  a multi-core runtime.
\newblock In {\em PPoPP}, 2006.

\bibitem{DBLP:journals/tocs/SavageBNSA97}
S.~Savage, M.~Burrows, G.~Nelson, P.~Sobalvarro, and T.~E. Anderson.
\newblock Eraser: A dynamic data race detector for multithreaded programs.
\newblock {\em Theor.\ Comp.\ Sci.}, 15(4), 1997.

\bibitem{schwoon}
S.~Schwoon.
\newblock {\em Model-Checking Pushdown Systems}.
\newblock PhD thesis, TUM, 2002.

\bibitem{DBLP:conf/popl/ShapiroH97}
M.~Shapiro and S.~Horwitz.
\newblock Fast and accurate flow-insensitive points-to analysis.
\newblock In {\em POPL}, 1997.

\bibitem{kn:SP81}
M.~Sharir and A.~Pnueli.
\newblock Two approaches to interprocedural data flow analysis.
\newblock In {\em Program Flow Analysis: {T}heory and Applications}, chapter~7.
  Prentice-Hall, Englewood Cliffs, NJ, 1981.

\bibitem{shavit:sw-trans-mem:podc:1995}
N.~Shavit and D.~Touitou.
\newblock Software transactional memory.
\newblock In {\em PODC}, 1995.

\bibitem{DBLP:conf/popl/Steensgaard96}
B.~Steensgaard.
\newblock Points-to analysis in almost linear time.
\newblock In {\em POPL}, 1996.

\bibitem{gems:website}
{Wisconsin {Multifacet} {GEMS} {Simulator}}.
\newblock http://www.cs.wisc.edu/gems/.

\bibitem{woo:splash2:isca:1995}
S.~C. Woo, M.~Ohara, E.~Torrie, J.~P. Singh, and A.~Gupta.
\newblock The splash-2 programs: Characterization and methodological
  considerations.
\newblock In {\em ISCA}, 1995.

\end{thebibliography}

\section{XOrder}
\label{Se:XOrder}
XOrder is an analysis that determines an improved layout for the
fields of a record type. It identifies these reorderings by analyzing the
results of XRef analysis. The experiments reported in \sectref{exp:xorder},
show that, without programmer intervention, XOrder can reduce the rate of
false conflicts for programs running on a transactional memory system.
\begin{definition}
\label{De:falseconflict}
A false conflict occurs when two distinct dynamic transactions access
separate bytes of memory, but, due to the conflict-detection scheme of
the transactional memory system, one of them is forced to stall or abort.
\end{definition}
Although false conflicts can occur in both STMs and HTMs, they are
especially prevelent in HTMs that detect conflicts using the coherence
mechanism. For clarity, in this section, we assume an HTM.

\begin{figure}
\begin{center}
\ifpdf
\includegraphics[width=35mm,height=25mm]{Images/FalseConflict.jpg}
\else
\includegraphics[width=35mm,height=25mm]{Images/FalseConflict.eps}
\fi
\end{center}
\caption{Possible layout in memory of the shared record {\tt s}
from \figref{Example}. Each row represents one cache line of memory.}
\label{Fi:FalseConflict}
\end{figure}

In our target HTM implementation, the granularity of precision is one cache
line. Therefore, during a dynamic transaction, the HTM's conflict-detection
mechanism
does not distinguish between memory reads and writes to the same cache line.
\figref{FalseConflict} depicts a possible layout in memory of the shared
{\tt ColoredShape} record \texttt{s} from the program in \figref{Example}. Assuming that
integers occupy four bytes of memory and that the cache-line size is sixteen
bytes, accesses to any of the fields \{\texttt{r, g, b, x1}\}
will be treated equally by the HTM's conflict-detection mechanism. Similarly,
accesses to any of \{\texttt{y1, x2, y2, x3}\} will not be distinguished and
accesses to any of \{\texttt{y3, x4, y4}\} will not be distinguished.
This memory alignment, combined with the HTM's
conflict-detection mechanism, will cause a false conflict
between the concurrent execution of transactions
({\tt n2},{\tt n3}) and ({\tt n1},{\tt n4}) from the program in
\figref{Example}. It is a false conflict because
the two transactions do not access any common fields of the shared
\texttt{ColoredShape} record \texttt{s}. To tackle this problem, we developed an
analysis algorithm, XOrder, that determines an optimized
layout ordering for record types. 

Record-layout reordering is a technique that analyzes which fields of
a particular record type might be accessed in a dynanic
transaction. The result of XRef is the set of fields of record types
accessed in a static transaction.  Thus, XRef computes a superset of
the fields actually accessed in a dynamic transaction. The idea behind
record-layout reordering is that if two fields, $f_1$ and $f_2$, are
accessed only in separate dynamic transactions, then the layout of the
record type in memory should be chosen to ensure that these fields
reside in separate cache lines. Because we are performing static
analysis, we do not know where in memory a record instance will be
dynamically created, and furthermore we do not know its alignment with
respect to cache lines.  Therefore, our goals are: (1) to separate
$f_1$ and $f_2$ with fields that are not accessed in a static
transaction, and (2) to ensure that the sum of the sizes of these
``padding'' fields is greater than or equal to the size of a cache
line. A beneficial byproduct of this ordering is that the number of cache
lines accessed by a dynamic transaction is potentially decreased.  This is
akin to the {\em cache-block working set} optimization described by Chilimbi
et al. in \cite{CDL99}.


XOrder uses the field annotations inferred by XRef to determine an optimized
ordering of a record type's fields. With respect to \cite{CDL99}, we do not
use dynamic profiling information to determine a field ordering. Instead, we
rely on the field accesses that occur in a static transaction.

Intuitively, the ordering determined by XOrder should be such that each static
transaction accesses a contiguous region of memory. When the fields accessed
by a static transaction do not overlap with the fields accessed by any other
static transaction, determining this ordering is trivial.
We create an undirected graph, where the nodes of the graph correspond to the
fields of the record type. There is an edge between two nodes (fields) if they
are accessed in the same static transaction. In \figref{Graph}, the {\tt x}
fields are all connected because they are accessed in the static transaction
({\tt n1},{\tt n4}) and the {\tt y} fields are all connected because they are
accessed in the static transaction ({\tt n2},{\tt n3}).
When ordering these fields, each connected component represents nodes that
should be laid out near each other. A (linearization of a) spanning tree for the
connected component gives a linear order for those fields.

In some cases, the accesses of two static transactions will share some subset
of a record type's fields. We refer to these static transactions as a
transactional group.
\begin{definition}
\label{De:TransactionalGroup}
A transactional group consists of 2 or more static transactions such that
there is a common field shared between at least two of the static transactions
in the group.
\end{definition}
In these situations,
%finding {\em any} spanning tree is not enough to
%determine the desired ordering. That is, finding a spanning tree for the
%transactional group and then linearizing the tree may result
%in an ordering that can still cause false conflicts.
%To combat this, we order the fields of each transactional group such that the
%fields common to multiple static transactions are placed near each other. This
%can reduce false conflicts when all the fields associated with a transactional
%group do not fit within one cache line. This is because fields that are unique
%to one static transaction will be as far away from the unique fields of other
%static transactions. For example, if a field common to two static transactions
%of a transactional group is a large array, placing this array between the
%fields not common to the static transactions of the transactional group could
%separate these fields by a cache line.\footnote{The model programming language
%used in this paper does not include arrays; however, our implementation is able to handle {\tt C}
%programs (including ones that use arrays).}
%
%Intuitively, the fields of a transactional group should be ordered such that
the fields of a transactional group are ordered such that
each static transaction of the group accesses contiguous regions of memory.
One solution to find such an ordering (if it exists) is to solve an instance
of the traveling-salesman problem (TSP). Create a graph where
the nodes of the graph represent the fields of the transactional group. For
each pair of nodes, add a weighted edge to the graph. The weight of the
edge is determined by the static transactions of the group. That is, if the
two fields represented by the nodes are accessed in the same static
transaction, then the weight is 1. Otherwise, we set the weight to be greater
than 1. By setting the weight to be greater than 1, a solution to the TSP
will not traverse that edge unless it is absolutely necessary. In our
experiments, we used the weight 10. For such graphs, an optimal solution to
the TSP will produce the desired ordering. The path can be used to create the
record layout. If it is possible for every static transaction of the group to
access contiguous memory, then this ordering will provide it. In cases where
no such ordering exists, the ordering produced by solving the generated
instance of the TSP will provide a good approximation.

Because we are targeting our analysis for a compiler infrastructure, solving
an NP-complete problem like the TSP every time the program is modified might
be too costly. Therefore, we use the well-known heuristic of using a
minimum-spanning-tree (MST) algorithm to approximate the TSP \cite{CLRS}.
Approximating the TSP solution using an MST is nice because it still takes the
graph's edge weights into account. For example, one could imagine 
using dynamic-profiling information to adjust the weights on the graph,
effectively biasing the solution to the TSP to prefer certain paths
(orderings) over others. This would allow XOrder to perform an optimization
akin to the {\em field-layout-affinity} method proposed by Chilimbi et al.
\cite{CDL99}. Because we use an MST to approximate a solution to the TSP, we no
longer need to distinguish between transactional groups and static
transactions (that are not members of a group). This is because we can use a
linearization of the MST to compute an ordering for both cases.

\begin{figure}
\begin{center}
\MediumScale\VCDraw{%
\begin{VCPicture}{(-4,-2)(4,2)}
\ShowFrame
\ShowGrid
%\ChgStateFillStatus{vlines}
\ChgStateLineStyle{dashed}
\State[x_1]{(-3.5,1)}{x1}
\State[x_2]{(-1.5,1)}{x2}
\State[x_3]{(-3.5,-1)}{x3}
\State[x_4]{(-1.5,-1)}{x4}

\ChgStateLineStyle{dotted}
%\ChgStateFillStatus{hlines}
\State[y_1]{(3.5,1)}{y1}
\State[y_2]{(1.5,1)}{y2}
\State[y_3]{(3.5,-1)}{y3}
\State[y_4]{(1.5,-1)}{y4}

\RstStateLineStyle
\State[r]{(0,1.5)}{r}
\State[g]{(0,0)}{g}
\State[b]{(0,-1.5)}{b}

\ChgEdgeArrowStyle{-}

\EdgeL{x1}{x2}{}
\EdgeL{x1}{x3}{}
\EdgeL{x1}{x4}{}
\EdgeL{x2}{x3}{}
\EdgeL{x2}{x4}{}
\EdgeL{x3}{x4}{}

\EdgeL{y1}{y2}{}
\EdgeL{y1}{y3}{}
\EdgeL{y1}{y4}{}
\EdgeL{y2}{y3}{}
\EdgeL{y2}{y4}{}
\EdgeL{y3}{y4}{}

\end{VCPicture}}

%\ifpdf
%\includegraphics[width=35mm,height=25mm]{Images/Graph.jpg}
%\else
%\includegraphics[width=40mm,height=30mm]{corr3.eps}
%\fi
\end{center}
\caption{The graph generated during XOrder analysis for the program in
\figref{Example}.}
\label{Fi:Graph}
\end{figure}

Once the layout of each static transaction and transactional group is
determined, we next use the record type's fields \emph{not} accessed in a static
transaction as padding between these orderings. If there are $n$ ordered sets,
then XOrder attempts to find $n-1$ sets of padding fields such that the size
of each set is at least the size of the cache line for the target
architecture. To do this, we sort the padding fields in decreasing order by
their size and then (attempt to) greedily fill the $n-1$ sets.
\figref{Reorder} displays the layout of  
\texttt{ColoredShape} after analyzing the program in \figref{Example} with
XOrder. Notice that the analysis suggested the insertion of an integer field
named {\tt pad1}. This is because for the example program 
(and target architecture with a cache-line size of 16 bytes),
there are not
enough fields to place between the two static transactions. Our analysis only
suggests the padding in a comment instead of directly emitting it because the
addition of extra fields can impact the performance of a program. For example,
if the program allocated a million {\tt ColoredShape} objects, then the
addition of the padding field would cause 4 MB of memory to be wasted.
Additionally, adding pad fields will affect the runtime behavior of the
program because it places extra pressure on the cache. Because of the padding
suggestions, we do not directly modify the analyzed program. Instead, we emit
the reorderings for each record type to the programmer. The programmer can
then decide whether to apply the reordering.
\input{CSquareReorder}

\subsection{Discussion}
\label{Se:XOrderDiscussion}
XOrder analysis is a whole program analysis. Field reordering can break
programs that make use of shared libraries that are compiled assuming a
certain field layout.  In addition to this, some programs write 
data to a persistent store. Reordering of fields can cause an
incompatibility with the data in the persistent store. However, if the
benefits from field reordering are large enough, we imagine the programming
effort required to make use of these fields is justifiable.

\subsection{XOrder Experiments}
\label{Se:exp:xorder}

\input{Benchmarks} We evaluated XOrder on three programs from the
SPLASH-2 benchmark suite \cite{woo:splash2:isca:1995}, modified to use
transactions in place of locks \cite{moore:logtm:hpca:2006}. The
SPLASH-2 programs are written in {\tt C}.\footnote{The {\tt C}
programming language is neither typesafe nor memory safe. Because of
this, both XRef and XOrder are unsound. The application of XRef and
XOrder analyses to a type and memory safe language would be sound.}
%We use the weighted pushdown system library WALi \cite{WALi}
%as our dataflow analysis solver.
%%%%% TODO: NAK rework this in?
%We extended CIL \cite{DBLP:conf/cc/NeculaMRW02} to extract a WPDS from the
%\texttt{C} code. 
%Our dataflow transformers are represented with BDDs, using
%the BuDDy \cite{buddy} BDD library.
%The analysis results returned from WALi
%are then fed into XOrder's record-layout engine.
%XOrder's record layout engine used the {\tt ocamlgraph} library
%\cite{ConchonFilliatreSignoles05} for all graph manipulations.
We modeled a 32-processor SPARC multiprocessor
running Solaris 9 with support for HTM using the Simics full-system
simulator \cite{magnusson:simics:ieeecomputer:2002} extended with a
detailed memory system timing model from the GEMS
\cite{martin:gems:can:2005,gems:website} toolkit.  Our simulated
system includes 32, 1 GHz SPARC processors each with 4 MB of private
cache.  The system includes HTM support based on LogTM
\cite{moore:logtm:hpca:2006}, which performs conflict detection at the
cache-line granularity.  For our experiments, we used a cache line
size of 64 bytes.

We compare the performance of the programs in their original form and
the programs modified to use the orderings suggested by XOrder. For
each of our benchmarks, we run both version of the program on our
simulated machine with a range of thread counts.  \tableref{Summaries}
displays the effect of the reordering on the overall runtime of the
benchmarks and on the frequency of transaction aborts, a major
contriubtor to the execution time.

Not surprisingly, our results vary for each benchmark. For Raytrace,
despite having little effect on the number of aborts, the reordering
provided by XOrder results in a dramatic improvement in performance (a
factor of 2.8 for 32 threads).  This result matches Moore et al.'s
observation that Raytrace suffers from false conflicts.  The
performance improvement is primarily due to the improved scalability
of the benchmark.  As shown in \figref{raytraceSpeedup}, eliminating
the false conflicts between transactions allows the reordered Raytrace
to continue to scale to 32 threads, while the original program
performance degrades after 16.

On Radiosity, XOrder at first appeared to provide an improvement. The number
of aborts steadily decreased up to 16 processors. However, However, the number
of aborts increased and performance worsened for 32 processors. We are
currently modifying the simulator to track additional processor state, e.g, we
are not able to separately measure true and false conflicts at this time. This
will help shed insight on the cause of the performance loss.

%performance was
%improve performance.seems to have little effect, improving
%performance for 16 threads, but degrading peformance for 32. We expect
%% FIXME
%that many workloads will not be improved by XOrder as they exhibit
%little to no false conflicts.  

For MP3D, on 16 processors, the layout XOrder suggested caused an increase in
the number of aborts which degraded performance for 16 threads. For 32
threads, despite a small increase in aborts, performance is unaffected.

Although our measured results show improvement in only one of our test cases,
we believe XOrder can still be a help to programmers.  XOrder found the
performance bug in Raytrace immediately. For MP3D, it was already tuned to
avoid false sharing.  Our future work includes analyzing more benchmarks and
further investigating the behavior of Radiosity. Hopefully, the investigations
can help us improve XOrder's field layout algorithm.
